Author's response to reviews

Title: Efficacy of Rifabutin-based Triple Therapy as Second-line to Eradicate Helicobacter Pylori Infection

Authors:

Jose M Navarro-Jarabo (jnavarro@hcs.es)
Nuria Fernandez (jnavarro@hcs.es)
Francisca L Sousa (jarabo@tiscali.es)
Encarnacion Cabrera (jarabo@tiscali.es)
Manuel Castro (jarabo@tiscali.es)
Luz M Ramirez (lmramirez@yahoo.es)
Robin Rivera (robinriverai@yahoo.es)
Esther Ubina (estherillaua@hotmail.com)
Francisco Vera (jarabo@tiscali.es)
Isabel Mendez (chabeli_estepona@hotmail.com)
Francisco Rivas-Ruiz (frivasar@hcs.es)
Jose L Moreno (jlmharo@hcs.es)
Emilio Perea-Milla (eperea@hcs.es)

Version: 6  Date: 5 July 2007

Author's response to reviews:

The following phrase has been modified:

"All patients signed an informed consent form"

by:

"All patients signed a written informed consent form"